Author's response to reviews

Title: Linear accelerator-based stereotactic radiosurgery in 140 brain metastases from malignant melanoma

Authors:

Henrik Hauswald (henrik.hauswald@med.uni-heidelberg.de)
Alina Stenke (alina.stenke@gmail.com)
Juergen Debus (juergen.debus@med.uni-heidelberg.de)
Stephanie E Combs (stephanie.combs@med.uni-heidelberg.de)

Version: 3 Date: 21 October 2014

Author's response to reviews:

I have revised the manuscript according to the guidelines. Following points were fixed:
- Abbreviations section added at end of manuscript
- Deleted abbreviations under each Table.
- No more Abbreviations in abstract.
- Fixed some abbreviations in the introduction
- Informed consent statement added to stats section.
- Reference number of ethics committee decision added.
- 1 reference fixed.
- Competing interests, authors contributions, funding now before references instead of title page.
- Now double line spacing instead of 1.5 before.

I am really sorry for these errors and hope I have addressed all of them.